**Author's response to reviews**

**Title:** Polysaccharides from the root of Angelica sinesis promotes hematopoiesis and thrombopoiesis through the PI3K/AKT pathway

**Authors:**

Mo Yang (yangm1091@yahoo.com.hk)
Chang Liu (cliu6688@yahoo.com)
Ruixia Deng (ruixia_deng@yahoo.com.cn)
Jieyu Ye (jieyu.ye@googlemail.com)
Jianqin Li (jqlcglff@126.com)
Haixia Yan (yanhaixia7993@yahoo.com.cn)

**Version:** 3  **Date:** 16 November 2010

**Author's response to reviews:**

Dear Editors:

We appreciate very much the reviewers’ prompt comments. Below are our point-by-point responses.

Referee 1:

Many in-vivo and in-vitro data are confusing for the reader. This article should be divided into two parts: one in-vivo and one in-vitro. Both parts could be separate articles. The actions on TLR or PI3K/AKT are too speculative.

Response: we appreciate the reviewer’s comments and will take them into account in our future manuscript. However, these comments lack specifics that can lead us to modify our manuscript.

Referee 2:

I am delighted that the authors have rewritten this paper. It is far, far better than the earlier version. Indeed, the authors’ need to be congratulated on the clarity of the presentation. I no longer have any concerns.

Response: we thank the reviewer for the positive feedback.

We made no change in the manuscript as no specifics are found in the comments. There is one technical problem as I can not change the author order through the submission site. We are looking forward to hearing from you regarding the decision on our manuscript.

Best regard,

Chang Liu